Author's response to reviews

Title: The BpTRU automatic blood pressure monitor compared to 24 hour ambulatory blood pressure monitoring in the assessment of blood pressure in patients with hypertension

Authors:

Linda Beckett (beckettfamily@cogeco.ca)
Marshall Godwin (godwinm@post.queensu.ca)

Version: 4 Date: 20 June 2005

Author's response to reviews:

Full stop removed from end of title in manuscript and submission system

Country - Country name included at end of the affiliation detail

Abstract- Elaborated on Background section in manuscript and submission page.

References - References cited in the text using consecutive numbers in square brackets.

References - all authors' names used in references.